Author’s response to reviews

Title: Single-stage transforaminal decompression, debridement, interbody fusion, and posterior instrumentation for lumbosacral brucellosis

Authors:
yakefu abulizi (yakup0213@126.com)
weidong liang (362746037@qq.com)
aikeremujiang muheremu (yakupspine@163.com)
weibin sheng (wbsheng@vip.sina.com)
Maierdan Maimaiti (halida0222@163.com)

Version: 3 Date: 08 Jun 2017

Author’s response to reviews:
Thank you for your letter for our manuscript entitled “Single-stage transforaminal decompression, debridement, interbody fusion, and posterior instrumentation for lumbosacral brucellosis” (BSUR-D-17-00077).

1. Response to comment: amend the “consent for publication” statement
Response: We have corrected the "consent for publication" statement according to the editor’s suggestion. Revised portion are marked red colors in the paper. (Consent for publication section, line 1, page 9)

2. Response to comment: gender and age
Response: All participants gave consent to reveal personal information, including the gender and age. we have considered that it is essential to show the correlation between the disease and characteristics of the patients. so we think it may be not necessary to change the table 1

3. Response to comment: Any items which are not cited may be deleted by our production department upon publication
Response: we agree with the any changes of our figures.